Author's response to reviews

Title: Inapplicability of advanced directives in a paternalistic setting: the case of a post-communist health system

Authors:

Gentian Vyshka (gvyshka@yahoo.com)
Jera Kruja (jkuja@gmail.com)

Version: 2 Date: 9 January 2011

Author's response to reviews: see over
Inapplicability of advanced directives in a paternalistic setting: the case of a post-communist health system

Gentian Vyshka* (1), Jera Kruja (2)

(1) Biomedical & Experimental Department, Faculty of Medicine, Rr. Dibrës 371, Tirana, Albania.

(2) Service of Neurology, University Hospital Centre “Mother Theresa”, Tirana, Albania.

Corresponding author: gyvshka@yahoo.com

The paper has been partially rewritten and re-organized; new references have been added.

Particular focus has been made in correcting grammatical errors and omissions; a second author, whose contribution was tangential in the first version, is added due to his complete mentoring of the writing process.

Abstract as well has been corrected and partially re-written. All corrections are anyway of a formal character, aiming at improving the quality of written text; thus the opinions expressed herein from the authors remain widely the same of the first version.

___________________________________________